

Author's response to reviews

Title: Changes in Macular Sensitivity after Half-Dose Photodynamic Therapy for Chronic Central Serous Chorioretinopathy

Authors:

Alp Atik (ala374@mail.harvard.edu)

Yijun Hu (huyijun2014@163.com)

Honghua Yu (yhhgtm@126.com)

Chun Yang (602757305@qq.com)

Bin Cai (105455762@qq.com)

Yijing Tao (taotao2212@hotmail.com)

Dongli Li (kyxiaoyi51@163.com)

Yan Chen (3206512935@qq.com)

Li Lu (872898493@qq.com)

Guodong Li (daliguodong@sina.com)

Ling Yuan (yuanling8061@163.com)

Version: 2 Date: 29 Jul 2017

Author's response to reviews:

1. Thank you for providing an Availability of data and materials subheading in your Declarations section. Please describe where the raw data generated/analysed during your study can be found (including information found in any supplementary files or repositories). If you do not wish to share your data, please state this along with the reason. If the data can be requested, please state so in this section.

Response: We have discussed with all the patients but most of them refused to share their raw data. We have added this information in the Declarations section (line 360-361).

2. When submitting your revised manuscript please ensure you do so as a clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

Response: The manuscript has been revised accordingly.